Author's response to reviews

Title: A cross-sectional analysis of the relationship between tobacco and alcohol outlet density and neighbourhood deprivation

Authors:

Niamh K Shortt (niamh.shortt@ed.ac.uk)
Catherine Tisch (catherine.tisch@ed.ac.uk)
Jamie Pearce (Jamie.pearce@ed.ac.uk)
Richard Mitchell (Richard.Mitchell@glasgow.ac.uk)
Elizabeth A Richardson (E.Richardson@ed.ac.uk)
Sarah Hill (S.E.Hill@ed.ac.uk)
Jeff Collin (jeff.collin@ed.ac.uk)

Version: 4 Date: 14 September 2015

Author's response to reviews:

Dear Editor

Ref: MS 2854490401692786 - A cross-sectional analysis of the relationship between tobacco and alcohol outlet density and neighbourhood deprivation

Many thanks for your email dates September 14th 2015. I respond to your points below:

1. I have added this Figure Legend list after the references. For some references I have just inserted a figure heading and for others I have inserted a heading and a legend.

2. I have permission, under the End User Licence Agreement (EULA) from Digimap UK to use and publish maps from the Ordnance Survey data. The following is taken from their FAQs: You are permitted to publish maps in various publications as long as no revenue is generated by you while conducting this activity. This is specified in clause 3.3.8 of the EULA.

3. I have added contributions from all authors in the contributions statement.

Best wishes

Niamh K Shortt